Author’s response to reviews

Title: Feasibility and impact of an evidence-based electronic decision support system. Diabetes care in family medicine as a case study: Protocol for a cluster randomized controlled trial.

Authors:

Annie Heselmans (annemie.heselmans@med.kuleuven.be)
Stijn Van de Velde (stijn.vandevelde@med.kuleuven.be)
Dirk Ramaekers (dirk.ramaekers@med.kuleuven.be)
Robert Vander Stichele (robert.vanderstichele@ugent.be)
Bert Aertgeerts (bert.aertgeerts@med.kuleuven.be)

Version: 2 Date: 3 July 2013

Author’s response to reviews:

Dear editor,

Thank you for reviewing the manuscript. The following adaptations were made:
- A native English speaker was contacted to proofread the text
- The CONSORT checklist was completed

Yours sincerely,

Annemie Heselmans